Geophysical Research Letters^{*}

RESEARCH LETTER

10.1029/2022GL099529

Key Points:

- Atmosphere-snow exchange of water vapor is a key process for the signal formation of the snow isotopic composition
- Taking into account isotopic fractionation during sublimation we explain a significant share of the observed day-to-day isotope variability
- Atmosphere-snow exchange introduces a warm-bias in the net summer snow isotope signal that is not accounted for in ice core interpretations

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

S. Wahl, sonja.wahl@uib.no

Citation:

Wahl, S., Steen-Larsen, H. C., Hughes, A. G., Dietrich, L. J., Zuhr, A., Behrens, M., et al. (2022). Atmosphere-snow exchange explains surface snow isotope variability. *Geophysical Research Letters*, 49, e2022GL099529. https://doi. org/10.1029/2022GL099529

Received 14 MAY 2022 Accepted 9 SEP 2022

Author Contributions:

Conceptualization: S. Wahl, H. C. Steen-Larsen Data curation: S. Wahl, H. C. Steen-Larsen, A. Zuhr, M. Behrens, M. Hörhold Formal analysis: S. Wahl Funding acquisition: H. C. Steen-Larsen Investigation: S. Wahl, H. C. Steen-Larsen, A. G. Hughes, L. J. Dietrich Methodology: S. Wahl, H. C. Steen-Larsen, A.-K. Faber Project Administration: S. Wahl, H. C. Steen-Larsen Software: S. Wahl, A. G. Hughes

© 2022. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Atmosphere-Snow Exchange Explains Surface Snow Isotope Variability

S. Wahl¹, H. C. Steen-Larsen¹, A. G. Hughes², L. J. Dietrich¹, A. Zuhr^{3,4}, M. Behrens⁵, A.-K. Faber¹, and M. Hörhold⁵

¹University of Bergen and Bjerknes Centre for Climate Research, Bergen, Norway, ²Institute of Arctic and Alpine Research, University of Colorado, Boulder, CO, USA, ³Alfred-Wegener-Institut Helmholtz Zentrum für Polar- und Meeresforschung, Research Unit Potsdam, Potsdam, Germany, ⁴University of Potsdam, Institute of Geosciences, Potsdam, Germany, ⁵Alfred-Wegener-Institut Helmholtz Zentrum für Polar- und Meeresforschung, Research Unit Bremerhaven, Bremerhaven, Germany

Abstract The climate signal imprinted in the snow isotopic composition allows to infer past climate variability from ice core stable water isotope records. The concurrent evolution of vapor and surface snow isotopic composition between precipitation events indicates that post-depositional atmosphere-snow humidity exchange influences the snow and hence the ice core isotope signal. To date, however, this is not accounted for in paeleoclimate reconstructions from isotope records. Here we show that vapor-snow exchange explains 36% of the summertime day-to-day δ^{18} O variability of the surface snow between precipitation events, and 53% of the δ D variability. Through observations from the Greenland Ice Sheet and accompanying modeling we demonstrate that vapor-snow exchange introduces a warm bias on the summertime snow isotope value relevant for ice core records. In case of long-term variability in atmosphere-snow exchange the relevance for the ice core signal is also variable and thus paleoclimate reconstructions from isotope records should be revisited.

Plain Language Summary Ice core water isotope records are valuable climate proxies containing information about past climate. Climate reconstructions from ice cores have been based on the interpretation of the water isotope records as precipitation signal. Here we show that the humidity exchange between atmosphere and snow influences the surface snow isotopic composition in-between precipitation events in summer. Thus, it is not the precipitation signal alone that defines the summer snow isotope signal but a combination of precipitation and post-depositional atmospheric vapor-snow exchange. This suggests, that climate reconstructions from ice core isotope records should account for post-depositional processes in the interpretation of the water isotope signal, specifically when interpreting summer isotope signals.

1. Introduction

Ice cores allow paleo-climate reconstructions through climate proxies such as stable water isotope records. An accurate understanding of the transfer function between the climatic conditions and the ice core stable water isotope signal is thus crucial and important for integrating stable water isotope records in climate models. The rationale behind using ice core isotope records as climate proxies is the temperature dependency of the isotopic composition of precipitation that can be described by a Rayleigh distillation process. However, the assumption that the ice core climate signal is governed by the precipitation isotope signal alone has previously been challenged (Casado et al., 2021; Steen-Larsen et al., 2014).

Besides uneven deposition through precipitation intermittency (Casado et al., 2020; Münch et al., 2021; Persson et al., 2011) and precipitation seasonality (Cuffey & Steig, 1998; Zheng et al., 2018), wind redistribution (Zuhr et al., 2021) and mixing with drifting snow (Stenni et al., 2016), post-depositional processes are suggested to alter the isotopic composition of the snow. Specifically, vapor-snow exchange processes occur continuously at the surface through turbulent humidity fluxes (Wahl et al., 2021) which are defined by local climate and below the surface during snow metamorphism. The latter is influenced by local climate through temperature shifts that induce temperature gradients in the snow and trigger interstitial vapor diffusion and associated snow recrystal-lization (Casado et al., 2021; Ebner et al., 2017; Touzeau et al., 2017). Thus, post-depositional processes represent a potential mechanism behind the observed strong correlation between local temperature and the isotopic composition of snow and ice on various timescales other than Rayleigh distillation. In-situ observations at ice



ഹ



Validation: S. Wahl Visualization: S. Wahl Writing – original draft: S. Wahl Writing – review & editing: S. Wahl, H. C. Steen-Larsen, L. J. Dietrich, A. Zuhr, M. Behrens, A.-K. Faber, M. Hörhold core sites in both Greenland and Antarctica documented strong covariances of snow and vapor isotopic composition between precipitation events (Ritter et al., 2016; Steen-Larsen et al., 2014). Furthermore, recent work demonstrated that isotopic fractionation during atmospheric vapor-snow exchange drives the observed diurnal cycle in the near surface vapor isotopic composition (Madsen et al., 2019). Previously, such isotopic fractionation during sublimation, and hence influences of sublimation on the snow isotopic composition, have been disregarded as influential process due to low diffusivities of water molecules in the solid snow crystal matrix, which prevent mixing within the solid phase (Dansgaard, 1964; Friedman et al., 1991). However, laboratory and field experiments demonstrated isotopic enrichment of snow during sublimation (Hughes, Wahl, et al., 2021; Stichler et al., 2001), and in-situ measurements of the sublimation flux's isotopic composition provided direct evidence for fractionation during sublimation (Wahl et al., 2021). The questions that remain include: (a) to which extent atmosphere-snow isotope exchange processes are modifying the original precipitated snow isotope signal before the signal is buried and archived in ice records and, additionally, (b) which environmental factors are controlling the magnitude of the overprinting of the precipitation's original signal. In case of a significant fraction of the buried isotope signal stemming from such post-depositional surface exchange processes, there is a need for revisiting existing paleoclimate reconstructions from ice core isotope records.

Here we address the role of atmospheric vapor-snow exchange of water isotopes for the surface snow isotopic composition in polar areas. We make use of summertime observations obtained within the interior dry snow zone of the Greenland Ice Sheet (GrIS) that we perceive representative for other low accumulation sites. We focus on the effect of vapor-snow exchange in the form of turbulent humidity fluxes during precipitation- and snowdrift-free periods in summer and evaluate the impact on the summer seasonal isotope signal; the period in which we expect highest sublimation-induced fractionation. To this end, a snow surface model is designed that is driven by in-situ meteorological, snow and vapor observations with the objective to simulate the observed evolution of the snow surface isotopic composition and to identify the responsible drivers.

2. Materials and Methods

2.1. Field Site

Field observations were obtained at the East Greenland Ice Core Project (EastGRIP) field site on the GrIS (75°37'47''N, 35°59'22''W, 2700 m a.s.l in 2019). The estimated local accumulation rate is 10–15 cm water equivalent (w. eq.) year⁻¹ (Karlsson et al., 2020). Measurements were taken in a dedicated clean-snow area upwind from the main camp facilities. The data set presented here entails data of two summer campaigns, one between 11 May–5 August 2018, the other in the period 17 May–31 July 2019.

2.1.1. Observations of Atmospheric Parameters

Continuous water vapor levels and corresponding stable water isotope ratios were measured simultaneously with a cavity ring-down laser absorption spectrometer (CRDS) (Picarro L2140-i). In 2018 (2019), the CRDS measured vapor at 1.8 m (2 m) above the snow surface for 15 min (30 min) every hour, and cycled through other measurement levels in the remaining time. Between 12:00 10 June and 21:00 1 July 2018 UTC the CRDS constantly measured at the 1.8 m level. Measurements were reported with ~1 Hz frequency and given in volume mixing ratio (ppmv) for humidity and in δ -notation (given %) for H₂¹⁸O (δ ¹⁸O) and ¹H ²HO (δ D) following Craig (1961):

$$\delta^* = \frac{R^*}{R^*_{\rm VSMOW}} - 1 \tag{1}$$

Where R^* is the abundance ratio of rare (* = ¹⁸O or ²H) to abundant (¹⁶O or ¹H) isotope and R^*_{VSMOW} is the respective Vienna Standard Mean Ocean Water abundance ratio.

The CRDS measurements were calibrated for humidity dependency and referenced against the VSMOW-SLAP scale (Text S1 in Supporting Information S1).

For the 2018 vapor data set 10 min periods were averaged yielding one measurement point per hour and a continuous 10 min timeseries between 10 June and 1 July. The 2019 data set contains 9 min averages amounting to three consecutive datapoints per hour. The resulting uncertainty on the averaged individual values is estimated to be 0.23% and 1.4% for δ^{18} O and δ D, respectively (Steen-Larsen et al., 2013).

As a measure for vapor-snow exchange, surface humidity fluxes were measured as latent heat fluxes (*LE*) using an eddy-covariance (EC) system (Text S2 in Supporting Information S1) oriented against the prevailing wind direction of ~240° (Wahl et al., 2021). The 1 standard deviation (SD) uncertainty on the *LE* measurement was conservatively estimated to be 20% (Litt et al., 2015; Vickers et al., 2010).

2.1.2. Observations of Snow Parameters

For estimating the isotopic composition of the surface snow, consolidated daily snow samples were collected from eleven 10 m-spaced and decorrelated (Stuart et al., 2021) locations along a 100 m long transect oriented in the prevailing wind direction. Per day, one sample was collected of the upper 0–0.5, 0–1, and 0–2 cm snow layer in 2018, and of the upper 0–0.5, 0–1, 0–2 and 0–5 cm in 2019 (Figure S1 in Supporting Information S1). The sample locations were marked to prevent collection of disturbed snow on the next day. The snow was sampled using an aluminum spatula, collecting equal amounts of snow at each location and transferring it to a plastic Whirl-Pak® in which it was kept frozen throughout storage and transport. The top layer snow samples from 16 July 2018 and 23 July 2019 showed signs of evaporation during transport and were excluded from the analysis. The snow samples were melted in the closed Whirl-Paks® to avoid evaporation and analyzed for their isotopic composition following van Geldern & Barth (2012) on a CRDS type Picarro L2140-i at the stable water isotope laboratory of the Alfred-Wegener-Institute in Bremerhaven, Germany. The measurement accuracy on the individual sample of 2018 (2019) is 0.07% (0.06%) and 0.11% (0.69%) for δ^{18} O and δ D, respectively as estimated as average offset of a quality control standard. In addition to the daily snow samples, a high-resolution (3 h) data set of snow samples (from three depths spanning 0-4 cm) for three periods of 2-3 days in 2019 is available (Hughes, Wahl, et al., 2021). Complementary to the isotope samples, snow density was measured at 10 locations along a different transect oriented perpendicular to the main wind direction by weighing snow samples of a known area and 2.5 cm thickness. Snow sampling and density measurements were performed roughly at the same time and thus share the same time information.

2.2. The Snow Surface Model

We use a numerical mass-balance snow surface model (Figure S2 in Supporting Information S1) to simulate the evolution of the top (0–0.5 cm) snow layer's isotopic composition that we compare against observations. The model consists of a layered 5 cm snowpack (4 cm for the high-resolution modeling). The vertical resolution is 0.5 cm with uniform snow density and temperature. At the time of a snow sampling event all model layers are initialized with the observed isotope values. For 2018, the deduced values of the 1–2 cm layers are used as initial conditions in the lowest 4 cm as, in contrast to 2019, no information is available of the snow isotopic composition below 2 cm. The snowpack's isotopic composition is then modulated by influences of surface fluxes affecting the top layer and isotope diffusion (Johnsen et al., 2000) affecting all layers. We use observations of meteorological and snow parameters as model forcing and reinitialized the model at the time of the next snow sampling event.

The time step of the model is 30 min, corresponding to the resolution of the *LE* observations. At every time step, an amount of snow *dm* (derived from *LE*) with a calculated isotopic composition (δ^*_F) is added (*LE* < 0) or removed (*LE* > 0) from the snowpack which modifies the isotopic composition of the top layer and the height of the snowpack.

During vapor deposition (LE < 0), δ_F^* is calculated from the 2 m vapor isotopic composition (δ_V^*) using the empirically derived linear equations of Wahl et al. (2021) to avoid assumptions of the effective supersaturation.

$$\delta_F^{18_0} = 1.50 \cdot \delta_F^{18_0} + 23.78\%$$
(2)

$$\delta_{F}^{D} = 2.24 \cdot \delta_{v}^{D} + 445.20\%$$
(3)

During sublimation (LE > 0), δ_F^* is calculated depending on the experimental set up. Three different experiments are performed in which we vary the nature of the fractionation process during sublimation by implementing (i) kinetic fractionation with a Craig and Gordon (1965) (CG) approach, (ii) equilibrium fractionation



 $(R_F^* = \alpha_{eq}^* (T_S)^{-1} \cdot R_S^*)$, or (iii) no fractionation ($\delta_F = \delta_S$). For experiment (i) we parameterize the kinetic fractionation and calculate δ_F^* following Merlivat and Jouzel's (1979) formulation of CG:

$$R_F^* = \frac{(1-k^*)}{(1-h)} \left(\frac{1}{\alpha_{eq}^*(T_S)} R_S^* - h R_V^* \right)$$
(4)

$$h = \frac{q_{2m}}{q_{\text{sat}}(T_S)} \tag{5}$$

Where α_{eq} are ice-vapor equilibrium fractionation coefficients from Majoube (1971) and Merlivat & Nief (1967), $k^{18} = 6\%$ and $k^D = 0.88 \cdot k^{18}$ are the kinetic fractionation factors, R_s^* is the isotopic composition of the top snow layer, Ts is the surface temperature and h is the humidity (q_{2m}) deficit compared to the saturation humidity level at the surface (q_{sat}) . Ts and relative humidity (h) at each time step are calculated from LE, q_{2m} and the measured friction velocity following the bulk flux theory (Van AS, 2011). By making h and T_s dependent on LE, we reduce the uncertainty associated with observations of the surface parameters and ensure that the model input parameters satisfy basic physical principles.

The change in the top snow layer isotopic composition due to vapor-exchange is calculated using a mass balance approach with a rate of change of

$$\frac{\partial \delta_S^*}{\partial t} = \frac{s}{m} \left(\delta_S^* - \delta_F^* \right) \tag{6}$$

mwhere *s* is the sublimation rate calculated from *LE*, *t* is time and *m* is the mass of the top surface layer. For a full model description see Text S5 of the Supporting Information S1. We run the model for the periods 24 May–3 August 2018 and 23 May–29 July 2019 in which we have overlapping atmospheric and snow parameter observations. We compute the net change in δ -values between daily sampling events ($\Delta\delta^*$) and compare the model results ($\Delta\delta^*_{model}$) against the observed changes ($\Delta\delta^*_{obs}$). We compute the ratio of explained variance (*VE*) from the 1:1 line as:

$$VE = 1 - \frac{\sum \left(\Delta \delta_{obs}^* - \Delta \delta_{model}^*\right)^2}{\sum \left(\Delta \delta_{obs}^* - \overline{\Delta \delta_{obs}^*}\right)^2}$$
(7)

Of the 158 (84 in 2018/74 in 2019) consolidated snow surface samples collected, 48 (30/18) samples were influenced by snowfall and are therefore excluded from the analyses. Likewise, 20 modeling periods in which max wind speeds of $>10 \text{ ms}^{-1}$ were observed were excluded to prevent influences from drifting snow. In the remaining model periods, the maximum 30 min averaged wind speed did not exceed 5.5 ms⁻¹, which is in compliance with literature thresholds for blowing snow (Birnbaum et al., 2010).

2.2.1. Sensitivity Analysis

We identify the most influential environmental drivers for the model results by performing a sensitivity analysis. For this we varied the input parameters δ_{v} , *LE* and the initial snow isotopic composition by ± 1 SD and ± 2 SD, where SD is the uncertainty estimate (Table S2 in Supporting Information S1), and evaluated changes in RMSE and *VE*. Additionally, the kinetic fractionation coefficient *k* was varied and the uncertainty in the sampling time accounted for by adding ± 2 h (in 30 min steps). We then calculate the SD in the set of resulting $\Delta\delta^*$ for each tested parameter and calculate the median of all modeling periods. We rank the model's sensitivity to the input parameters by the resulting SD of the medians. The results are given in Table S2 and Figure S4 in Supporting Information S1.

2.2.2. Uncertainty Estimates of Model Result

For estimating the uncertainty of the modeled $\Delta\delta^*$, we performed 1,000 Monte Carlo simulations for each day-today modeling period by adding random noise to the input parameters *LE*, δ_v , initial δ_s and atmospheric humidity *q*. The resulting distribution's SD is taken as modeling uncertainty and visualized as error bars in Figure 3. Text S3 in the Supporting Information S1 explains the estimated uncertainty level for each model parameter.





Figure 1. Observations of seasonal snow surface and vapor isotope variability. Shown are timeseries of the δ_S^{18} O signal of daily consolidated samples (purple circles) and the daily 2 m vapor isotope signal δ_V^{18} O (solid blue line) in the two observation periods in 2018 (a) and 2019 (b). Days with recorded snowfall are grayed out. The red vertical lines frame the modeling periods.

3. Results

3.1. Vapor and Snow Interplay

The daily consolidated surface (0–0.5 cm) snow samples show covarying temporal evolution with daily mean 2 m vapor isotopic composition (Figure 1). The snow isotopic composition (δ_s) of the two observation periods was distinct, with more enriched values in 2019 compared to 2018 (Table S1 in Supporting Information S1). Similarly, the 2 m water vapor (δ_v) was more enriched in 2019 compared to the 2018 summer season. The snow surface isotope signal was more variable over time with a higher standard deviation in 2019 compared to 2018.

Substantial changes in the isotopic composition of the snow corresponding to synoptic weather regime changes were recorded not only during, but also between precipitation events. Both isotopic enrichment and depletion events spanning several days were observed.

Throughout both summer observation periods, the recorded humidity flux amounted to a net sublimation with \sim 3 mm (5 mm sublimation and 2 mm deposition) w. eq. in 2018 and \sim 5 mm (7 mm sublimation and 2 mm deposition) w. eq. in 2019.

3.2. Snow Model Performance on Sub-Daily Timescales

To quantify the role of atmospheric vapor-snow exchange for the snow isotope signal, we simulate the evolution of the snow surface isotopic composition at the field site. First, we use the high-resolution snow data set (observations of snow surface isotopic composition every 3 hours) to investigate the model performance on sub-daily timescales and thus to evaluate sublimation and deposition periods separately.

Figure 2 shows that the model is able to capture the observed snow evolution on sub-daily timescales qualitatively during both snow isotopic enrichment and depletion. The pearson correlation coefficients (r) between modeled and observed values are $r \ge 0.83$ for all three periods. However, isotopic enrichment during sublimation (LE > 0) is slightly overestimated compared to observations. Despite the simple empirical linear relationship used for the deposition conditions, the modeled isotopic depletion during deposition periods (LE < 0) is consistent with observations.





Figure 2. Simulations and observations of sub-daily snow surface isotope variability. The simulated evolution (small red triangles) and the compared values (bigger red triangles) of the δ_s signal during the high-resolution snow sampling periods are shown. The observations (purple triangles, error bars show SD between samples from three locations) serve as model validation. The model is reinitialized at every observation point and run until the next available observation point. The forcing parameter *LE* with deposition (hatched) and sublimation (shaded) phases is shown in gray. The 2 m vapor isotopic composition (δ_v) is shown as blue line. Pearson correlation coefficients for a, b, and c are r = 0.83, r = 0.86, r = 0.83, respectively. Time is UTC.

3.3. Modeling of Snow Surface Day-to-Day Isotope Variability

Next, we make use of the full span of simultaneous vapor and snow isotope measurements of the two field campaigns and simulate the change of the snow isotopic composition in between daily snow sampling events, which will be referred to as day-to-day variability. We compare our modeled day-to-day variability of snow isotopic composition to the in-situ observed changes. Results of the experiment including kinetic fractionation (i) are shown in Figure 3 for $\Delta\delta^{18}$ O and $\Delta\delta$ D in (a) and (b), respectively. A one-to-one line is shown to indicate model performance. Based on the *VE* calculation (Equation 7), the model explains 36% of the variance in δ^{18} O observed in the top snow layer on days without snowfall and wind drift and 53% of the δ D variance. Both model and observations show that higher changes in snow surface isotopic composition coincide with higher net sublimation.

When excluding both equilibrium and kinetic fractionation during sublimation and allowing for diffusion and deposition processes only (iii), the VE values for both isotopic species approach zero (Figures 3c and 3d). If





Figure 3. Modeled versus observed day-to-day change in snow isotopic composition. The modeled day-to-day change in snow isotopic composition (δ^{18} O in (a) and δ D in (b)) using kinetic fractionation (experiment i) is plotted against the observed day-to-day change. The simulation with no fractionation during sublimation (experiment iii) is shown for δ^{18} O and δ D in (c) and (d), respectively. All model periods (N = 53) are color coded according to the cumulative *LE* observed between the respective two sampling events, bracketing the model period. 1-to-1 lines are shown in all plots. The uncertainties on the observations are inferred as square-summed standard error of the mean whereas the uncertainties on the model results were estimated using a Monte Carlo approach.

equilibrium fractionation (ii) is used instead of a CG type model the VE values are 0.30 for δ^{18} O, and 0.25 for δ D, respectively (Figure S3 in Supporting Information S1). See Table S2 in Supporting Information S1 for an overview of the performance metrics for all experiments and an experiment without isotope diffusion.

3.3.1. Uncertainty Evaluation

The model result ($\Delta \delta_s$) is most sensitive to uncertainties in the initial snow isotopic composition, followed by *LE*, δ_v and the timing of snow sampling (Figure S4 in Supporting Information S1). Model sensitivity for normalized relative humidity (*h*) and surface temperature (*Ts*) were not tested individually, since they are calculated from *LE*. Generally, variability in the perturbation response was higher for periods with higher observed $\Delta \delta_s$ values. Hence the relative importance of vapor-snow exchange processes on the snow isotopic composition varies throughout the season. Figure 2 shows that the surface snow model slightly overestimates the enrichment due to sublimation. The overestimation is counteracted by reduced *LE* or shortened modeling periods and explains the increased *VE* in the respective sensitivity model runs (Table S2 in Supporting Information S1). Similarly, the model performs better when vapor or snow enrichment. Approaching equilibrium, the gradient between snow and vapor isotopic composition is reduced, which reduces the effect of kinetic fractionation in the snow.

4. Discussion

4.1. Evaluation of the Model Performance

To draw conclusions about the dominant processes at play, we compare the experiments in terms of VE values and spread of data points around the one-to-one lines. We find that experiment (i), including kinetic fractionation during sublimation, performed best albeit overestimating the isotopic enrichment. Experiment (ii) with equilibrium fractionation performed moderately but underestimated the enrichment, wheras experiment (iii) without fractionation during sublimation performed worst. The improved model performance for δD compared to δ^{18} O in (i) can be explained by the increased sensitivity of δ^{18} O to kinetic effects (Table S2 in Supporting Information S1). The model results do not change significantly when isotope diffusion is disabled (Table S2 in Supporting Information S1) which demonstrates that the key process responsible for the model performance is the magnitude of isotopic fractionation during sublimation. However, the overestimation in experiment (i) suggests that the CG approach might be sub-optimal when simulating the fractionation process during snow sublimation. One way to improve model fidelity might be the introduction of a temperature-dependency which impacts the magnitude of fractionation. It has been suggested that temperature controls whether the diffusion within the solid snow matrix is effective enough to allow homogenization of the remaining snow crystal and thus controls the imprint of the fractionation signal (Hu et al., 2022). In other words, temperature can control snow metamorphism and thus impacts the fractionation strength. Additionally, ambient temperature shifts can establish snow temperature gradients and thus induce sub-surface vapor fluxes from below the 5 cm model domain (Casado et al., 2021) that can affect the surface layer. Whether it is reasonable to include a temperature dependency when simulating sublimation-fractionation will be subject of future studies.

4.2. Climate Signal Formation Through Vapor-Snow Exchange

Our results show that by incorporating kinetic fractionation during sublimation, the snow surface model is able to explain a substantial amount of the to-date unexplained observed variability in the surface snow isotope signal. By adapting the CG type sublimation model, we allow for feedbacks from the vapor isotopic composition on the magnitude of sublimation effects (Gat, 1996). Since the vapor isotopic composition is modulated by temperature-dependent Rayleigh distillation and synoptic weather events (Bagheri Dastgerdi et al., 2021; Steen-Larsen et al., 2013) it is a climate proxy itself. Vapor-snow exchange constitutes a pathway for how the atmospheric climate signal in the vapor isotopic composition can be imprinted on the snow isotope signal during precipitation-free periods. During sublimation it is the gradient between vapor and snow isotopic composition that, together with the sublimation rate, defines the magnitude of sublimation effects in the snow, which is visible in the sensitivity analysis of this study. Thus, we argue it is this isotopic gradient, that is, the disequilibrium between the actual vapor signal and vapor in equilibrium with the snow, that is the main driver for the flux isotopic composition (Benson et al., 1994), and is consequently important for the magnitude and direction of changes in the snow isotope signal. During deposition the vapor isotopic composition directly defines the deposition flux isotope signal and thus affects the snow signal (Casado et al., 2016; Wahl et al., 2021).

4.3. The Effect of Sublimation for the Summer Isotope Signal

For evaluating the importance of sublimation for the summer signal in the snowpack isotope record the comparison of the precipitation versus the sublimation signal both in terms of mass balance and preserved signal seems necessary. However, with no information about the precipitation's isotopic composition and mass (Zuhr et al., 2021), this comparison is infeasible. Instead, we calculate the observed change in snow isotopic composition during the observation periods and exclude days with snowfall or wind redistribution. The observations show a net enrichment in snow isotopic composition during fresh snow-free periods in both 2018 and 2019 (Table S1 in Supporting Information S1). Performing the same calculation on the no fractionation simulation (iii) data set yields a net depletion over the observation period due to deposition and diffusion influences (Table S1 in Supporting Information S1). The offsets between both calculations of ~14‰ and ~6‰ for δ^{18} O in 2018 and 2019, respectively demonstrate that sublimation significantly enriches the snow that is in contact with the atmosphere. Note that the calculated values can not be interpreted as the winter-summer amplitude in the δ^{18} O signal which is observed in the range of 10–15‰ (Komuro et al., 2021). Instead, it is the sublimation contribution to the summer signal, which is overlaid by variability emanating from precipitation. However, it demonstrates that vapor-snow exchange at EastGRIP introduces a warm-bias in the summer isotope signal through sublimation and therefore affects the annual isotope value.

4.4. Significance of Vapor-Snow Exchange on Longer Timescales

Whether vapor-exchange processes imprint on the snowpack on seasonal to annual timescales thus depends on the local reoccurrence of precipitation events, that is, the length of precipitation free periods, the magnitude and net direction of the surface humidity flux and the disequilibrium between vapor and snow surface. Since those climate characteristics are likely variable both spatially and temporally, the importance of vapor-snow exchange on the snow isotopic composition is not constant. Our findings suggest that variability in ice core isotope records could include a vapor-snow exchange component rather than a precipitation signal alone. This is particularly relevant for seasonal-amplitude climate studies but also for studies of longer time scales when the overall magnitude of the humidity flux changes, for example, during glacial and interglacial periods. Hence paleoclimate interpretations of ice core isotope records (e.g., Kawamura et al., 2017; Masson-Delmotte et al., 2005; Steffensen et al., 2008) should be revisited in the context of a warm-bias introducing vapor-snow exchange.

5. Conclusions

We investigated the summertime snow isotope variability on the GrIS that is the foundation for ice core isotope records which are interpreted as climate proxies. We set up a simple snow model of the upper 5 cm of the snow column and forced it with observations of surface fluxes, snow and the isotope signal of atmospheric water vapor. By modeling the evolution of the snow surface isotope signal we investigated the exchange of water isotopes between the snow and atmosphere as a snow isotope signal formation process. Our results showed that by incorporating kinetic fractionation during sublimation we can explain a substantial amount of the day-to-day variability that is observed in the surface snow isotopic composition in between precipitation events. In contrast, when omitting isotopic fractionation during sublimation the model fidelity decreased drastically which supports existing studies of a fractionation-inducing sublimation process. A comparison between modeled and observed net change of the snow isotope signal over the season demonstrated that atmospheric vapor-snow exchange has an enriching effect on the summer snow that can be attributed to sublimation, that is, sublimation can introduce a warm-bias in climate reconstruction studies when ice core water isotope records are interpreted without accounting for vapor-snow exchange. Considering that atmosphere-snow exchange processes are not constant in time, the impact on the snow isotope signal has likely varied in the past. Thus we suggest to revisit climate reconstruction studies from ice core stable water isotope records in the framework of a summer snow isotope signal that is partly defined by atmosphere-snow exchange.

Data Availability Statement

The observational isotopic and meteorological data sets used as input for the numerical model and as validation reference in the study are available in the PANGAEA data repository under the CC-BY licence. Data sets of the water vapor isotope measurements and surface fluxes are available as Steen-Larsen et al. (2022) and Steen-Larsen and Wahl (2021a, 2021b, 2022a). The snow isotope data is available as Hörhold et al. (2022a, 2022b) and Hughes, Steen-Larsen, et al. (2021). The snow density and snow temperature datasets are available as Steen-Larsen et al. (2022a) and Steen-Larsen and Wahl (2022b). A basic version of the snow surface model is published on Zenodo at https://doi.org/10.5281/zenodo.7117642.

References

Bagheri Dastgerdi, S., Behrens, M., Bonne, J.-L., Hörhold, M., Lohmann, G., Schlosser, E., & Werner, M. (2021). Continuous monitoring of surface water vapour isotopic compositions at Neumayer Station III, East Antarctica. *The Cryosphere*, 15(10), 4745–4767. https://doi. org/10.5194/tc-15-4745-2021

Benson, L. V., Hostetler, S. W., & Benson, L. V. (1994). Stable isotopes of oxygen and hydrogen in the Truckee River-Pyramid Lake surface-water system. *Limnology & Oceanography*, 39(2), 344–355. https://doi.org/10.4319/lo.1994.39.8.1945

Birnbaum, G., Freitag, J., Brauner, R., Kö Nig-Langlo, G., Schulz, E., Kipfstuhl, S., et al. (2010). Strong-wind events and their influence on the formation of snow dunes: Observations from Kohnen station, Dronning Maud Land, Antarctica. *Journal of Glaciology*, 56(199), 891–902. https://doi.org/10.3189/002214310794457272

Acknowledgments

The authors are grateful for the contstructive feedback from Mathieu Casado and one anonymous reviewer that improved this manuscript. Further, the authors would like to thank all EastGRIP participants who helped in snow sampling. This research has received funding from the European Research Council (ERC). European Union's Horizon 2020 research and innovation program: Starting Grant SNOWISO (Grant agreement. 759526) (recipient HCSL). EastGRIP is directed and organized by the Centre for Ice and Climate at the Niels Bohr Institute, University of Copenhagen. It is supported by funding agencies and institutions in Denmark (A. P. Møller Foundation, University of Copenhagen), USA (US National Science Foundation, Office of Polar Programs), Germany (Alfred Wegener Institute, Helmholtz Centre for Polar and Marine Research), Japan (National Institute of Polar Research and Arctic Challenge for Sustainability), Norway (University of Bergen and Trond Mohn Foundation), Switzerland (Swiss National Science Foundation), France (French Polar Institute Paul-Emile Victor, Institute for Geosciences and Environmental research). Canada (University of Manitoba) and China (Chinese Academy of Sciences and Beijing Normal University).

- Casado, M., Cauquoin, A., Landais, A., Israel, D., Orsi, A., Pangui, E., et al. (2016). Experimental determination and theoretical framework of kinetic fractionation at the water vapour-ice interface at low temperature. *Geochimica et Cosmochimica Acta*, 174, 54–69. https://doi.org/10.1016/j.gca.2015.11.009
- Casado, M., Landais, A., Picard, G., Arnaud, L., Dreossi, G., Stenni, B., & Prié, F. (2021). Water isotopic signature of surface snow metamorphism in Antarctica. *Geophysical Research Letters*, 48(17), 1–11. https://doi.org/10.1029/2021gl093382
- Casado, M., Münch, T., & Laepple, T. (2020). Climatic information archived in ice cores: Impact of intermittency and diffusion on the recorded isotopic signal in Antarctica. *Climate of the Past*, 16(4), 1581–1598. https://doi.org/10.5194/cp-16-1581-2020
- Craig, H. (1961). Standard for reporting concentrations of deuterium and oxygen-18 in natural waters. *Science*, 133(3467), 1833–1834. https://doi.org/10.1126/science.133.3467.1833
- Cuffey, K. M., & Steig, E. J. (1998). Isotopic diffusion in polar firm: Implications for interpretation of seasonal climate parameters in ice-core records, with emphasis on central Greenland. *Journal of Glaciology*, 44(147), 273–284. https://doi.org/10.1017/S0022143000002616
- Dansgaard, W. (1964). Stable isotopes in precipitation. Tellus, 16(4), 436-468. https://doi.org/10.3402/tellusa.v16i4.8993
- Ebner, P. P., Steen-Larsen, H. C., Stenni, B., Schneebeli, M., & Steinfeld, A. (2017). Experimental observation of transient δ¹⁸O interaction between snow and advective airflow under various temperature gradient conditions. *The Cryosphere*, 11(4), 1733–1743. https://doi. org/10.5194/tc-11-1733-2017
- Friedman, I., Benson, C., & Gleason, J. (1991). Isotopic changes during snow metaporphism. In H. P. Tayoir, J. R. O'Neill, & I. R. Kaplan (Eds.), Stable isotope geochemistry: A tribute to Samuel Epstein (Vol. 3, pp. 211–221). Geochemical Society. Retrieved from https://www.geochemsoc.org/files/8114/1269/7654/SP-3_211-222_Friedman.pdf
- Gat, J. R. (1996). Oxygen and hydrogen isotopes in the hydrologic cycle. Annual Review of Earth and Planetary Sciences, 24(1), 225–262. https://doi.org/10.1159/000088336
- Hörhold, M., Behrens, M., Wahl, S., Faber, A.-K., Zuhr, A., Meyer, H., & Steen-Larsen, H. C. (2022a). Snow stable water isotopes of a surface transect at the EastGRIP deep drilling site, summer season 2019. PANGAEA. https://doi.org/10.1594/PANGAEA.945563
- Hörhold, M., Behrens, M., Wahl, S., Faber, A.-K., Zuhr, A., Zolles, T., & Steen-Larsen, H. C. (2022b). Snow stable water isotopes of a surface transect at the EastGRIP deep drilling site, summer season 2018. PANGAEA. https://doi.org/10.1594/PANGAEA.945544
- Hu, J., Yan, Y., Yeung, L. Y., & Dee, S. G. (2022). Sublimation origin of negative deuterium excess observed in snow and ice samples from McMurdo dry valleys and Allan hills blue ice areas, East Antarctica. *Journal of Geophysical Research: Atmospheres*, 127(11), e2021JD035950. https://doi.org/10.1029/2021jd035950
- Hughes, A. G., Steen-Larsen, H. C., Wahl, S., & Jones, T. R. (2021). Stable water isotope data of snow samples and atmospheric vapor from laboratory experiments and field experiments (EastGRIP site, Greenland Ice Sheet summer 2019). PANGAEA. https://doi.org/10.1594/ PANGAEA.937355
- Hughes, A. G., Wahl, S., Jones, T. R., Zuhr, A., Hörhold, M., White, J. W. C., & Steen-Larsen, H. C. (2021). The role of sublimation as a driver of climate signals in the water isotope content of surface snow: Laboratory and field experimental results. *The Cryosphere*, 15(10), 4949–4974. https://doi.org/10.5194/tc-15-4949-2021
- Johnsen, S. J., Clausen, H. B., Cuffey, K. M., Hoffmann, G., Schwander, J., & Creyts, T. (2000). Diffusion of stable isotope in polar firn and ice: The isotope effect in firn diffusion. *Physics of Ice Core Records*. https://doi.org/10.7916/D8KW5D4X
- Karlsson, N. B., Razik, S., Hörhold, M., Winter, A., Steinhage, D., Binder, T., & Eisen, O. (2020). Surface accumulation in Northern Central Greenland during the last 300 years. Annals of Glaciology, 61(81), 214–224. https://doi.org/10.1017/aog.2020.30
- Kawamura, K., Abe-Ouchi, A., Motoyama, H., Ageta, Y., Aoki, S., Azuma, N., et al. (2017). State dependence of climatic instability over the past 720, 000 years from Antarctic ice cores and climate modeling. *Science Advances*, 3(2), e1600446. https://doi.org/10.1126/sciadv.1600446
- Komuro, Y., Nakazawa, F., Hirabayashi, M., Goto-Azuma, K., Nagatsuka, N., Shigeyama, W., et al. (2021). Temporal and spatial variabilities in surface mass balance at the EGRIP site, Greenland from 2009 to 2017. *Polar Science*, 27, 100568. https://doi.org/10.1016/j.polar.2020.100568
- Litt, M., Sicart, J. E., & Helgason, W. (2015). A study of turbulent fluxes and their measurement errors for different wind regimes over the tropical Zongo Glacier (16°S) during the dry season. Atmospheric Measurement Techniques, 8(8), 3229–3250. https://doi.org/10.5194/ amt-8-3229-2015
- Madsen, M. V., Steen-Larsen, H. C., Hörhold, M., Box, J., Berben, S. M. P., Capron, E., et al. (2019). Evidence of isotopic fractionation during vapor exchange between the atmosphere and the snow surface in Greenland. *Journal of Geophysical Research: Atmospheres*, 124(6), 2932– 2945. https://doi.org/10.1029/2018JD029619
- Majoube, M. (1971). Fractionnement en 18O entre la glace et la vapeur d'eau. Journal de Chimie Physique, 68(4), 625–636. https://doi.org/10.1051/jcp/1971680625
- Masson-Delmotte, V., Jouzel, J., Landais, A., Stievenard, M., Johnsen, S. J., White, J. W. C., et al. (2005). GRIP deuterium excess reveals rapid and orbital-scale changes in Greenland Moisture Origin. *Science*, 309(5731), 118–121. https://doi.org/10.1126/science.1108575
- Merlivat, L., & Nief, G. (1967). Fractionnement isotopique lors des changements d'etat solide-vapeur et liquide-vapeur de l'eau a des temperatures inferieures a 0°C. *Tellus*.
- Münch, T., Werner, M., & Laepple, T. (2021). How precipitation intermittency sets an optimal sampling distance for temperature reconstructions from Antarctic ice cores. *Climate of the Past*, 17(4), 1587–1605. https://doi.org/10.5194/cp-17-1587-2021
- Persson, A., Langen, P. L., Ditlevsen, P., & Vinther, B. M. (2011). The influence of precipitation weighting on interannual variability of stable water isotopes in Greenland. *Journal of Geophysical Research*, 116(D20), D20120. https://doi.org/10.1029/2010JD015517
- Ritter, F., Christian Steen-Larsen, H., Werner, M., Masson-Delmotte, V., Orsi, A., Behrens, M., et al. (2016). Isotopic exchange on the diurnal scale between near-surface snow and lower atmospheric water vapor at Kohnen station, East Antarctica. *The Cryosphere*, 10(4), 1647–1663. https://doi.org/10.5194/tc-10-1647-2016
- Steen-Larsen, H. C., Hörhold, M., Kipfstuhl, S., Faber, A.-K., Freitag, J., Hughes, A. G., et al. (2022a). 10 daily surface measurements over 90m transect, SSA, Density and Accumulation, from EastGRIP summer (May–August) of 2016-2019. PANGAEA. https://doi.org/10.1594/ PANGAEA.946763
- Steen-Larsen, H. C., Johnsen, S. J., Masson-Delmotte, V., Stenni, B., Risi, C., Sodemann, H., et al. (2013). Continuous monitoring of summer surface water vapor isotopic composition above the Greenland Ice Sheet. Atmospheric Chemistry and Physics, 13(9), 4815–4828. https://doi. org/10.5194/acp-13-4815-2013
- Steen-Larsen, H. C., Masson-Delmotte, V., Hirabayashi, M., Winkler, R., Satow, K., Prié, F., et al. (2014). What controls the isotopic composition of Greenland surface snow? *Climate of the Past*, 10(1), 377–392. https://doi.org/10.5194/cp-10-377-2014
- Steen-Larsen, H. C., & Wahl, S. (2021a). 2m processed sensible and latent heat flux, friction velocity and stability at EastGRIP site on Greenland Ice Sheet, summer 2019. PANGAEA. https://doi.org/10.1594/PANGAEA.928827
- Steen-Larsen, H. C., & Wahl, S. (2021b). Calibrated 2m stable water vapor isotope data from EastGRIP, Greenland Ice Sheet summer 2019. *PANGAEA*. https://doi.org/10.1594/PANGAEA.928823

- Steen-Larsen, H. C., & Wahl, S. (2022a). Calibrated 1.8 m stable water vapor isotope data from EastGRIP site on Greenland Ice Sheet, summer 2018. https://doi.org/10.1594/PANGAEA.946740
- Steen-Larsen, H. C., & Wahl, S. (2022b). Snow temperature at 10 cm depth, EGRIP 2019. https://doi.org/10.1594/PANGAEA.946730
- Steen-Larsen, H. C., Wahl, S., Box, J. E., & Hubbard, A. L. (2022). Processed sensible and latent heat flux, friction velocity and stability at EastGRIP site on Greenland Ice Sheet. https://doi.org/10.1594/PANGAEA.946741
- Steffensen, J. P., Andersen, K. K., Bigler, M., Clausen, H. B., Dahl-Jensen, D., Fischer, H., et al. (2008). High-resolution Greenland ice core data show abrupt climate change happens in few years. *Science*, 321(5889), 680–684. https://doi.org/10.1126/science.1157707
- Stenni, B., Scarchilli, C., Masson-Delmotte, V., Schlosser, E., Ciardini, V., Dreossi, G., et al. (2016). Three-year monitoring of stable isotopes of precipitation at Concordia Station, East Antarctica. *The Cryosphere*, 10(5), 2415–2428. https://doi.org/10.5194/tc-10-2415-2016
- Stichler, W., Schotterer, U., Fröhlich, K., Ginot, P., Kull, C., Gäggeler, H., & Pouyaud, B. (2001). Influence of sublimation on stable isotope records recovered from high-altitude glaciers in the tropical Andes. *Journal of Geophysical Research*, 106(D19), 22613–22620. https://doi. org/10.1029/2001JD900179
- Stuart, R. H., Faber, A., Wahl, S., Hörhold, M., Kipfstuhl, S., Vasskog, K., et al. (2021). Exploring the role of snow metamorphism on the isotopic composition of the surface snow at EastGRIP. *The Cryosphere Discussions*, 1–27. https://doi.org/10.5194/tc-2021-344
- Touzeau, A., Landais, A., Morin, S., Arnaud, L., & Picard, G. (2017). Numerical experiments on isotopic diffusion in polar snow and firm using a multi-layer energy balance model. *Geoscientific Model Development Discussions*, 1–58. https://doi.org/10.5194/gmd-2017-217
- van Geldern, R., & Barth, J. A. C. (2012). Optimization of instrument setup and post-run corrections for oxygen and hydrogen stable isotope measurements of water by isotope ratio infrared spectroscopy (IRIS). *Limnology and Oceanography: Methods*, 10(12), 1024–1036. https:// doi.org/10.4319/lom.2012.10.1024
- Van AS, D. (2011). Warming, glacier melt and surface energy budget from weather station observations in the Melville Bay region of northwest Greenland. *Journal of Glaciology*, 57(202), 208–220. https://doi.org/10.3189/002214311796405898
- Vickers, D., Göckede, M., & Law, B. E. (2010). Uncertainty estimates for 1-h averaged turbulence fluxes of carbon dioxide, latent heat and sensible heat. *Tellus Series B Chemical and Physical Meteorology*, 62(2), 87–99. https://doi.org/10.1111/j.1600-0889.2009.00449.x
- Wahl, S., Steen-Larsen, H. C., Reuder, J., & Hörhold, M. (2021). Quantifying the stable water isotopologue exchange between the snow surface and lower atmosphere by direct flux measurements. *Journal of Geophysical Research: Atmospheres*, 126(13), 1–24. https://doi. org/10.1029/2020JD034400
- Zheng, M., Sjolte, J., Adolphi, F., Møllesøe Vinther, B., Steen-Larsen, H. C., Popp, T. J., & Muscheler, R. (2018). Climate information preserved in seasonal water isotope at NEEM: Relationships with temperature, circulation and sea ice. *Climate of the Past*, 14(7), 1067–1078. https:// doi.org/10.5194/cp-14-1067-2018
- Zuhr, A. M., Münch, T., Steen-Larsen, H. C., Hörhold, M., & Laepple, T. (2021). Local-scale deposition of surface snow on the Greenland Ice Sheet. *The Cryosphere*, 15(10), 4873–4900. https://doi.org/10.5194/tc-15-4873-2021

References From the Supporting Information

- Andreas, E. L. (1987). A theory for the scalar roughness and the scalar transfer coefficients over snow and sea ice. *Boundary-Layer Meteorology*, 38(1), 159–184. https://doi.org/10.1007/BF00121562
- Craig, H., & Gordon, L. I. (1965). Deuterium and Oxygen 18 variations in the ocean and the marine atmosphere. In E. Tongiorgi (Ed.), *Stable isotopes in oceanographic studies and paleotemperatures* (pp. 9–130). Laboratorio di Geologia Nucleare.
- Goff, J. A., & Gratch, S. (1946). Low-pressure properties of water from -160 to 212°F (pp. 95–122). Transactions of the American Society of Heating and Ventilating Engineers.
- Holtslag, A. A. M., & De Bruin, H. A. R. (1988). Applied modeling of the nighttime surface energy balance over land. Journal of Applied Meteorology, 27(6), 689–704. https://doi.org/10.1175/1520-0450(1988)027<0689:AMOTNS>2.0.CO;2
- Mauder, M., Liebethal, C., Göckede, M., Leps, J.-P., Beyrich, F., & Foken, T. (2006). Processing and quality control of flux data during LITFASS-2003. Boundary-Layer Meteorology, 121(1), 67–88. https://doi.org/10.1007/s10546-006-9094-0
- Merlivat, L. (1978). Molecular diffusivities of H216O, HD16O, and H218O in gases. The Journal of Chemical Physics, 69(6), 2864–2871. https://doi.org/10.1063/1.436884
- Merlivat, L., & Jouzel, J. (1979). Global Climatic interpretation of the deuterium excess-oxygen 18 relationship for precipitation. Journal of Geophysical Research, 84(C8), 5029–5033. https://doi.org/10.1029/JC084iC08p05029
- Paulson, C. A. (1970). The mathematical representation of wind speed and temperature profiles in the unstable atmospheric surface layer. Journal of Applied Meteorology, 9(6), 857–861. https://doi.org/10.1175/1520-0450(1970)009<0857:TMROWS>2.0.CO;2
- Schwander, J., Stauffer, B., & Sigg, A. (1988). Air mixing in firn and the age of the air at pore close-off. Annals of Glaciology, 10, 141–145. https://doi.org/10.1017/S0260305500004328